Author’s response to reviews

Title: Social equity in Human Papillomavirus vaccination: A natural experiment in Calgary Canada

Authors:

Richard Musto (Richard.Musto@albertahealthservices.ca)
Jodi E. Siever (Jodi.Siever@albertahealthservices.ca)
Cyne Johnston (Cyne.Johnston@albertahealthservices.ca)
Judy Seidel (Judy.Seidel@albertahealthservices.ca)
Sarah Rose (rosesarah32@gmail.com)
Deborah A. McNeil (Debbie.Mcneil@albertahealthservices.ca)

Version: 3 Date: 19 June 2013

Author’s response to reviews: see over
Thank you for accepting our manuscript for publication in your journal. We have reviewed the manuscript according to the associate editor’s comments and made the following responses and changes to the manuscript.

**Associate Editor’s Comments:**

1. All tables and figures should specify time (when?), place (where?) and person (population in the study?, name of study?).

   We have included the following additional information in the title of each figure and each table: (Grade 5 and 9 Girls in Calgary Alberta, Canada, 2008-2011)